Author’s response to reviews

Title: Rapid Increase of Overweight and Obesity among Primary School-aged Children in the Caribbean; High Initial BMI is the most Significant Predictor

Authors:

Walaa Mumena (wmouminah@gmail.com)
Isabella Francis-Granderson (isabella.granderson@gmail.com)
Leroy Phillip (leroy.phillip@mcgill.ca)
Katherine Gray-Donald (katherine.gray-donald@mcgill.ca)

Version: 2 Date: 13 Jan 2018

Author’s response to reviews:

1. Please represent authors’ names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Done

2. Please edit the ethics section to include the full name of each of the ethics committees that provided approval for your study.

Done

3. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Done

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done